Tue Nov 19 17:17:05 GVI 1996
Year: 96 Doy: 324
observer: koon
WEATHER COMMVENT: Tue Nov 19 17:17:10 GMI 1996
Cl ear sky, wind=10 nph fromthe SSW tenp=42 F

Tue Nov 19 17:17:45 GMI' 1996 CH P Startup--Initializing new tape
Tue Nov 19 17:21:31 GMI 1996 dPMbn Start Patro
Tue Nov 19 17:21:41 GMI 1996 CH P CH P Start Patro

Tue Nov 19 18:00:55 GV 1996 dPMon Fl at
Tue Nov 19 18:01:49 GVIT 1996 dPMon End FIl at

Tue Nov 19 18:01:48 GVII 1996 CH P Bi as
Tue Nov 19 18:02:54 GVII 1996 CH P End Bi as
Tue Nov 19 18:03: 05 GVIT 1996 CH P Wt er
Tue Nov 19 18:03:45 GVIT 1996 CH P End Wat er

MKI 11 COMMENT: Tue Nov 19 18:17:03 GMI 1996
The DEC disk drive #2 Ready Light appears to be burnt out.
Tue Nov 19 19:01:03 GV 1996 dPMon Fl at

Tue Nov 19 19:01:50 GVII 1996 CH P Bi as

Tue Nov 19 19:01:55 GMI' 1996 dPMon End Fl at

Tue Nov 19 19:02:58 GVIT 1996 CH P End Bi as

Tue Nov 19 19:03:09 GVII 1996 CH P Wat er

Tue Nov 19 19:03:43 GVIT 1996 CH P End Wat er
Tue Nov 19 20:01:46 GVII 1996 CH P Gai n

Tue Nov 19 20: 02: 00 GMI 1996 dPMbn Fl at

Tue Nov 19 20: 02: 56 GMI 1996 dPMon End Fl at

MKI '] COMMENT: Tue Nov 19 20:16:21 GMI 1996

Repl aced the bulb for the DEC DD #2 Ready button.

Tue Nov 19 20: 31:37 GMI 1996 CH P End Gain

Tue Nov 19 20:31:47 GVI 1996 CH P Bi as

CH P COMMENT: Tue Nov 19 20:31:43 GMI 1996

Used Fi xbusy to clear an Optics:Busy stall that occurred during the gain
portion of the cal

CH P COMMENT: Tue Nov 19 20:47:31 GMI 1996

Pr ogram now hung up on the Bias portion of the cal, used Kill/Run to fix.
Tue Nov 19 21:01:57 GV 1996 dPMon Fl at

Tue Nov 19 21:02:50 GMI 1996 dPMon End FI at

Tue Nov 19 21:03:33 GVl 1996 CH P CH P Start Patro
MKI '] COMMENT: Tue Nov 19 21:18:07 GMI 1996
Latest scan shows no signal and no data are going to tape, will restart with

a new tape.

MKI'TI COWMMENT: Tue Nov 19 21:30: 15 GMI 1996

Still no signal after rebooting.

Tue Nov 19 22:01:48 GMI 1996 CH P Bi as
Tue Nov 19 22:01:58 GMI 1996 dPMon Fl at

Tue Nov 19 22:02: 50 GMTI 1996 dPMon End Fl at



Tue Nov 19 22:02:55 GMI' 1996 CH P End Bi as

Tue Nov 19 22:03: 05 GMI' 1996 CH P Wat er

Tue Nov 19 22:03:42 GMI 1996 CH P End Water

MKI '] COWMMENT: Tue Nov 19 22:18:51 GMI 1996

There is no signal fromeither channel #0 or #1. This appears to have started
around the time | was changing the Ready Lightbulb for DD #2, and Corona
was running while | did the change so nmaybe sonet hi ng happened when data
was supposed to go to DD#2 but couldn't locate it due to the assignnment key
bei ng renoved while | changed the bulb. But |ooking back through my notes
a simlar problem happened and was traced to the anal og boards, but before
could try swappi ng boards with known good boards the problemcl eared, so
maybe it is back again. Unfortunately | have to mail some things in Hilo
then get to an inportant appointnment so | can't do any troubl eshooting

today. | also tried turning off the conputer and anal og rack and restarting
but problemwas still there.

Tue Nov 19 22:38:03 GV 1996 dPMon End Patro

Tue Nov 19 22:40:11 GVI 1996 CH P CH P End Patro

Tue Nov 19 22:41:53 GVII 1996 CH P endi ng tape

COMMENT: Tue Nov 19 22:41:59 GMI 1996
Activity report:

QP: 45; 113; 122; 232-244; 313.

No coronal activity.

TAPES:

MKITT: HO1438

DPMON: P00764

CH P: C00181

LOAL: LO00427

SCAN- LOG

SCAN- LOG 17:22:04. 11/19/96 DOY 324
17:27: 01 17:30: 12 17: 33: 26 17: 36: 39 17:39: 53
17: 43: 06 17: 46: 21 17: 49: 36 17:52: 52 17: 56: 07
17:59: 25 18: 02: 42 18: 06: 03 18: 09: 19 18: 12: 36
1820 0 CL 18: 23: 44 1828 15CL 18: 32: 32 18: 40: 53
18: 44: 13 18: 47: 36 18: 50: 53 18:54: 13 18: 57: 32
19: 00: 51 19: 04: 09 19: 07: 28 19: 10: 46 19: 14: 04
19:17: 22 19: 20: 41 19: 23: 59 19: 27: 19 19: 30: 35
19: 33: 52 19: 37: 08 19: 40: 25 19: 43: 41 19: 46: 58
19: 50: 14 19:53: 31 19: 56: 46 20: 00: 03 20: 03: 18
20: 06: 34 20: 09: 49 20: 13: 06 20: 16: 21 20: 19: 39
20:19: 39 20:19: 39 20:19: 39 20:19: 39 20:19: 39
20:19: 39 20:19: 39 20:19: 39 20:19: 39 20:19: 39
20:19: 39 20:19: 39 20:19: 39 20:19: 39 20:19: 39

0 ERRORS

X



